\begin{table}
%\centering
\par
%\footnotesize
\caption{\label{tableA4}Additional parameters.}
\begin{tabular}{lrrrrrrrrr}
\hline
\hline
\noalign{\smallskip}
Star                     & $\log \dot{M}_{\rm acc}$ &  {\it EW}(H$\alpha$)  & {\it EW}([O~{\sc i}]$_f$) &   {\it EW}([O~{\sc i}]$_t$) & $\log L_{{\rm [OI]},_f}$  & $\log L_{{\rm [OI]},t}$  & $\log \dot{M}_{\rm loss}$    & jet? & Refs.\tablefootmark{b}   \\ 
                         & ($M_{\odot}$~yr$^{-1}$)  &      (\AA)      & (\AA)                &   (\AA)                & ($L_{\odot}$)             & ($L_{\odot}$)          & ($M_{\odot}$~yr$^{-1}$)   & TD?\tablefootmark{a}  &   \\
\hline
RNO~15                   &  ...   & ...    &     ...   &  ...   &   ...  &    ...  &    ...  &   ...  &  ...\\  
LkH$\alpha$~270          &  ...   &   30.9 &     ...   &  ...   &   ...  &    ...  &    ...  &   ...  &  6 \\  
LkH$\alpha$~271          &  ...   & 185.7  &      ...  &  ...   &   ...  &    ...  &    ...  &   ...  &  6\\  
LkH$\alpha$~326          &  ...   &   52.7 &     ...   & 0.8    &   ...  &    ...  &    ...  &   ...  &  6 \\  
LkH$\alpha$~327          &  ...   &   51.0 &     ...   & 0.9    &   ...  &    ...  &    ...  &   ...  &  6 \\  
LkH$\alpha$~330          &  ...   &   20.3 &     ...   &  ...   &   ...  &    ...  &    ...  &   T    &  6, 31 \\  
IRAS 03446+3254          &  ...   &  ...   &     ...   & ...    &   ...  &    ...  &    ...  &   ...  &  ... \\  
BP~Tau                   & --7.54  &  40--92 &    0.07   &   0.26 &  --5.71 & --4.75   & $<$--9.7 &   ...  &  13, X, 12 \\  
FM~Tau                   & --8.45  &  62--71 &     0.045 & 0.48   &  --6.37 &  --5.19  & $<$--10.6&   ...  &  21, 35, 6, 12       \\  
T~Tau N                  & --7.12  &  38--60 &     ...   &  2.0   &   ...  &  --2.78  &  --6.96  &   J    &  21, 6, 26, 4, (27) \\  
T~Tau S                  & ...    &   ...  &     ...   &  ...   &   ...  &    ...  &    ...  &   J    &  4 \\  
LkCa~8                   & --9.10  &  ...   &   0.24    & 0.63   & --5.3   &  --4.77  &   --9.6  &   ...  &  13, 12 \\  
DG~Tau                   & --6.30  &  63--125&    12.4   &   17.5 &  --2.21 & --2.07   &   --6.5  &   J    &  21, X, 12, 29 \\  
IQ~Tau                   & --7.55  &  8--17  &     ...   &  0.4   &   ...  &  --4.89  &    ...  &   ...  &  21, 6, (5, 27) \\  
FX~Tau                   & --8.65  &  9.6   &      ...  &  ...   &   ...  &    ...  &    ...  &   ...  &  21, 6\\  
DK~Tau                   & --7.42  &  19--50 &    0.36   &   0.79 &  --4.23 & --3.90   &   --8.5  &   ...  &  13, X, 12 \\  
V710~Tau                 & ...    &  34--89 &    ...    & 0.3    &   ...  &  --5.00  &    ...  &   ...  &  X, 6, (5, 27) \\  
GI~Tau                   & --8.02  &  15--21 &    0.11   &   0.48 &  --5.11 & --4.46   &   --9.4  &   ...  &  13, X, 12 \\  
GK~Tau                   & --8.19  &  15--35 &    0.20   &   0.51 &  --4.88 & --4.48   &   --9.2  &   ...  &  13, X, 12 \\  
HN~Tau                   & --8.60  &120--163 &     5.7   & 13.2   &  --3.82 &  --3.46  &  --8.1   &   J    &  35, 6, 12, 37   \\  
DM~Tau                   & --7.95  & 139    &    ...    &  ...   &   ...  &   ...   &    ...  &   T    &  13, 6, 33 \\  
AA~Tau                   & --8.48  &  37--46 &    0.16   &   1.25 &  --4.77 & --3.88   &   --9.1  &   J    &  13, X, 12, 30 \\  
DN~Tau                   & --8.46  &  12--87 &   0.1     & 0.53   &  --5.07 &  --4.34  &   --9.4  &   ...  &  21, X, 12 \\  
CoKu Tau 3               &  ...   &    5   &     ...   &  ...   &   ...  &    ...  &    ...  &   ...  &  6 \\  
DO~Tau                   & --6.84  & 100--109&    2.63   &   3.60 &  --3.20 & --3.06   &   --7.5  &   J    &  13, X, 12, 17 \\  
CoKu Tau 4               &$<$--10.0&   ...  &     ...   &  ...   &   ...  &    ...  &    ...  &   T    &  21, 34 \\  
DP~Tau                   & --7.88  & 74--102 &   0.7     &  0.7   &  --4.78 & --4.78   &  --7.42  &   J    &  13, X, 17, 26, 28, (27) \\  
UY~Aur                   & --7.18  &  36--73 &    1.15   &   2.30 &  --3.91 & --3.62   &   --8.2  &   J    &  13, X, 12, 17 \\  
GM~Aur                   & --8.02  &  96--97 &   0.015   &   0.27 &  --6.25 & --4.90   &$<$--10.0 &   T    &  13, X, 12, 33 \\  
V836~Tau                 & --7.86  &  25    &     0.02  & 0.35   &  --6.60 &  --4.94  &$<$--10.1 &   T    &  26, 12, 38  \\  
RW~Aur                   & --7.12  &  42-84 &     1.18  & 1.59   &  --3.26 &  --3.11  &  --7.6   &   J    &  21, 6, 26, 35, 12, 36     \\  
BF~Ori                   &  ...   &  6.4   &      ...  &  ...   &   ...  &    ...  &    ...  &   ...  &  6\\  
IRAS 08267-3336          &  ...   &  ...   &     ...   & ...    &   ...  &    ...  &    ...  &   ...  &  ... \\  
WX~Cha                   & --8.47  &  66    &      ...  &  ...   &   ...  &   ...   &    ...  &    ..  &  11, 9\\  
SX~Cha                   & --8.37  &  27    &      ...  &  ...   &   ...  &   ...   &    ...  &   ...  &  13, 9 \\  
SY~Cha                   & --8.60  &  8--64  &      ...  &  ...   &   ...  &    ...  &    ...  &   ...  &  13, 20\&9 \\  
TW~Cha                   &  ...   & 26--41  &      ...  & 0.95   &   ...  &  --3.84  &    ...  &   ...  &  20\&9, 10, (9)\\  
TW~Hya                   & --8.70  & 220    &    ...    & 0.67   &   ...  &  --5.06  &    ...  &   T    &  14, 40, 16, 32 \\  
CS~Cha                   & --7.92  &  13--65 &    ...    &  ...   &   ...  &   ...   &    ...  &   J, T  &  8, 9\&11, 25, 8 \\  
BYB~35                   &  ...   &    ... &  ...      &  ...   &   ...  &     ... &   ...   &   ...  &  ...\\  
CHXR~30                  &  ...   &  1     &      ...  & ...    &   ...  &   ...   &    ...  &  ...   &  20  \\  
VW~Cha                   & --6.95  &  71--147&      ...  & 0.93   &   ...  &  --3.63  &    ...  &   J    &  13, 20\&9, 10, 3, (9) \\  
VZ~Cha                   & --8.28  &    71.4&      ...  & 0.42   &   ...  &  --4.75  &    ...  &   ...  &  13, 9, 10, (9) \\  
ISO-ChaI~237             &  ...   & 3.8    &      ...  &  ...   &   ...  &   ...   &    ...  &    ..  &  20 \\  
Ced 111 IRS 7            & ...    & ...    &    ...    &  ...   &   ...  &    ...  &    ...  &   ...  &  ... \\  
HM~27                    & ...    & 102--201&       ... &    ... &     ...&     ... &   ...   &     ...&   20\&9\\  
XX~Cha                   & --9.07  & 90--134 &      ...  &  ...   &   ...  &    ...  &    ...  &   ...  &  13, 20\&9 \\  
RX~J1111.7-7620          & --9.30  &   7.7  &    ...    &  ...   &   ...  &    ...  &    ...  &    T   &  22, 20 \\  
T~Cha                    &  ...   &  2-7   &      ...  &  ...   &   ...  &    ...  &    ...  &    T   &  41, 31 \\  
IRAS~12535-7623          & ...    & ...    &       ... &    ... &     ...&     ... &    ...  &     ...&  ...\\  
Sz~50                    & ...    &   ...  &      ...  &  ...   &   ...  &    ...  &    ...  &    J   &  1 \\  
ISO-ChaII~54             &  ...   &  ...   &     ...   &  ...   &   ...  &    ...  &    ...  &   ...  &  ... \\  
DL~Cha                   & ...    & ...    &       ... &    ... &     ...&     ... &    ...  &   ...  &   ...\\  
HT~Lup                   & ...    &  2.8   &       ... & $<$0.03&  ...   & $<$--5.02&   ...   &   ...  &  18, 10, (M, 19)\\  
GW~Lup                   & ...    &  90.3  &    ...    &  ...   &   ...  &    ...  &    ...  &   ...  &  18 \\  
Sz~73                    &  ...   &   97.2 &      ...  &  ...   &   ...  &    ...  &    ...  &   ...  &  18 \\  
GQ~Lup                   & --7.50  &  38.6  &    ...    & 0.08   &   ...  &  --5.34  &    ...  &   ...  &  24, 18, 10, (M, 19) \\  
IM~Lup                   & ...    &    8.1 &     ...   & ...    &   ...  &    ...  &    ...  &   ...  &  18 \\  
RU~Lup                   & --7.30  &  216.4 &     0.79  &   1.94 &   --4.3 &  --3.95  &    ...  &   J    &  15, 18, 10, 25, (S, M, 15) \\  
RY~Lup                   & ...    & ...    &       ... &    ... &     ...&     ... &    ...  &   ...  &   ...\\  
EX~Lup                   & ...    & ...    &       ... &    ... &     ...&     ... &    ...  &   ...  &   ...\\  
Sz~102                   & --8.10  &  377.4 &      ...  &  ...   &   ...  &    ...  &    ...  &   J    &  7, 18, 2 \\  
\hline
AS~205                   & ...    & 154.6  &       ... &   1.8  &     ...&   3.16  &      ...&   ...  &   6, (M, 6) \\  
PZ99 J161411             & --9.50  & ...    &    ...    &  ...   &   ...  &    ...  &    ...  &   T    &  22 \\  
Haro 1-1                 &  ...   &  123   &     ...   & 0.52   &   ...  &  --4.62  &    ...  &   ...  &  6,10,(S,6) \\  
Haro 1-4                 &  ...   &   47.6 &     ...   & ...    &   ...  &    ...  &    ...  &   ...  &  6 \\  
DoAr~24E                 & --8.21  & ...    &       ... &    ... &     ...&     ... &      ...&   ...  &  39 \\  
DoAr~25                  &$<$--9.24&  ...   &      ...  & ...    &   ...  &   ...   &    ...  &  ...   &  39 \\  
SR~21                    &$<$--8.84& ...    &       ... &    ... &     ...&     ... &      ...&   T    &  39, 31 \\  
IRS~51                   &  ...   &  ...   &     ...   &  ...   &   ...  &    ...  &    ...  &   ...  &  ... \\  
SR~9                     & --8.26  &   11.6 &      ...  & 0.06   &   ...  &  --6.00  &    ...  &   ...  &  39, 6, 10, (S, 6) \\  
V853~Oph                 & --8.31  &    42  &      ...  &  ...   &   ...  &    ...  &    ...  &   ...  &  39, 23 \\
ROX~42C                  &  ...   &  ...   &     ...   &  ...   &   ...  &    ...  &    ...  &   ...  &  ... \\  
ROXs~43A                 & ...    & ...    &       ... &    ... &     ...&     ... &      ...&   ...  &  ...\\  
IRS~60                   &  ...   &  ...   &     ...   &  ...   &   ...  &    ...  &    ...  &   ...  &  ... \\  
Haro 1-16                &  ...   &   59   &     ...   & ...    &   ...  &    ...  &    ...  &   ...  &  6 \\  
Haro 1-17                &  ...   &   15   &    ...    &  ...   &   ...  &    ...  &    ...  &   ...  &  23 \\  
RNO~90                   &  ...   &  ...   &     ...   &  ...   &   ...  &    ...  &    ...  &   ...  &  ... \\  
Wa~Oph 6                 &  ...   &   ...  &      ...  &  ...   &   ...  &    ...  &    ...  &   ...  &  ... \\  
V1121~Oph                &  ...   &   ...  &      ...  &  ...   &   ...  &    ...  &    ...  &   ...  &  ... \\  
VV~Ser                   &  ...   &   ...  &         ..&  ...   &     ...&      ...&      ...&   ...  &  ...\\  
SSTc2d~J182900.8         &  ...   &    ... &    ...    &    ... &     ...&     ... &	 ... &   ...  &  ...\\  
SSTc2d~J182909.8         &  ...   &    ... &    ...    &    ... &     ...&     ... &	 ... &   ...  &  ...\\  
SSTc2d~J182928.2         &  ...   &  ...   &      ...  &  ...   &   ...  &    ...  &    ...  &   ...  &  ... \\  
EC~74                    &  ...   &  ...   &     ...   &  ...   &   ...  &    ...  &    ...  &   ...  &  ... \\  
EC~82                    &  ...   &  ...   &     ...   &  ...   &   ...  &    ...  &    ...  &   ...  &  ... \\  
EC~90                    &  ...   &    ... &    ...    &    ... &     ...&     ... &   ...   &   ...  &  ...\\  
EC~92                    &  ...   &  ...   &     ...   & ...    &   ...  &    ...  &    ...  &   ...  &  ... \\  
CK~4                     &  ...   &    ... &    ...    &    ... &     ...&       ..&   ...   &	 ...  &  ...\\  
LkH$\alpha$~348          &  ...   &    310 &    ...    &    0.6 &     ...&       ..&   ...   &	 ...  &  6  \\  
RX~J1842.9-3542          & --9.0   &    ... &    ...    &    ... &     ...&       ..&   ...   &	 T    &  22\\  
RX~J1852.3-3700          & --9.3   &    ... &    ...    &    ... &     ...&       ..&   ...   &	 T    &  22\\  
\noalign{\smallskip}\hline
\end{tabular} 
\tablefoot{
\tablefootmark{a}{T = transitional disk (either ``anemic'' or ``cold'' disk); J~=~jet-driving object.}
\tablefootmark{b}{References:
    1 \citet{alcala08}; 
    2 \citet{bacciotti99};
    3 \citet{bally06};
    4 \citet{boehm94};
    5 \citet{briceno02};
    6 \citet{cohen79};
    7 \citet{comeron03}; 
    8 \citet{espaillat07};
    9 \citet{gauvin92};
    10 \citet{hamann94};
    11 \citet{hartigan93};
    12 \citet{hartigan95};
    13 \citet{hartmann98};
    14 \citet{herczeg04};
    15 \citet{herczeg05};
    16 \citet{herczeg07};
    17 \citet{hirth97};
    18 \citet{hughes94};
    19 \citet{krautter97};
    20 \citet{luhman04};
    21 \citet{najita07};
    22 \citet{pascucci07};
    23 \citet{rydgren80};
    24 \citet{seperuelo08};
    25 \citet{takami03};
    26 \citet{white04};
    27 \citet{kenyon95};
    28 \citet{mundt98};
    29 \citet{solf93};
    30 \citet{cox05};
    31 \citet{brown07};
    32 \citet{calvet02};
    33 \citet{calvet05};
    34 \citet{forrest04};
    35 \citet{white01};
    36 \citet{dougados00};
    37~\citet{hartigan04};
    38 \citet{najita08};
    39 \citet{natta06};
    40 \citet{webb99};
    41 \citet{alcala93}.
    M: 2MASS; 
    S: SIMBAD; 
    X: XEST survey, see \citet{guedel07a} and references therein.
    References in parentheses are for complementary data, e.g., $A_{\rm V}$ or R magnitude, used to calculate $\log L_{\rm [OI]}$.}}
\end{table}